Author’s response to reviews

Title: A case report of scrub typhus complicated with myocarditis and rhabdomyolysis

Authors:

Young Jae Ki (bigpapi0427@chosun.ac.kr)
Dong-Min Kim (drongkim@chosun.ac.kr)
Na Ra Yoon (shine@chosun.ac.kr)
Sung Soo Kim (kholywater@chosun.ac.kr)
Choon Mee Kim (choonmee@chosun.ac.kr)

Version: 2 Date: 13 Oct 2018

Author’s response to reviews:

1. We note a figure and table are referred to in the text yet none are attached to your document. Please ensure that all figures and tables you refer to are included in your manuscript.

Reply: We attached all figures and tables

2. Please remove any information referring to attaining consent to publish from your “Ethics approval and consent to participate” section. Further please amend this statement to “Not applicable”. For this manuscript, we would not expect ethics approval to have been obtained as it is a case report.

Reply: We have edited the part of the text which you above mentioned. Please refer to revised manuscript (Ethics approval and consent to participate section, line 260, page 13)

3. Please include the following statement in your “Availability of data and materials” section: “All data generated or analysed during this study are included in this published article”.

Reply: We have edited the part of the text which you above mentioned. Please refer to revised manuscript (Availability of data and materials section, line 266, page 13)

4. Please remove the title “Conclusions” and rename the “Discussion” to “Discussion and conclusions” as per our submission guidelines (https://bmcinfectdis.biomedcentral.com/submission-guidelines/preparing-your-manuscript/case-report).
Reply: We have edited the part of the text which you above mentioned. Please refer to revised manuscript (Discussion and conclusions section, line 155, page 8)

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply: We have confirmed that there are nothing what you mentioned in our manuscript